CENWP-OP-B                                                                                                                          14 May 2007
MEMORANDUM FOR CENWP-OP, ATTN: Operations Biologist
SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for Week 19 of 2007, which covers the period from 06 through 12 May 2007.
1 OPERATION SUMMARY

1.1 Daily average river flows ranged from 261.5 to 285.2 kcfs.  Daily average project tailwater elevation ranged from 20.2’ to 21.5’ msl.  Daily average powerhouse forebay elevation ranged from 74.0’ to 75.7’ msl.  Daily average river temperature ranged between 53(F and 56(F.  Secchi disk measurements averaged 4.0’ to 5.5’.

1.2 Spill Bays 1 and 18, which supply attraction flow for Cascades Island and B-Branch respectively, are included in the spill pattern.  Daily average spill ranged from 92.5 to 97.5 kcfs.  
1.3 Unit Operation. Powerhouse 2 is the priority powerhouse.  
1.3.1 Unit 8 remains out of service for turbine rehab.
1.3.2 Unit 10 was out of service from 0300 on 07 May until 1518 on 10 May for slip ring adjustments.

1.3.3 Unit 11 returned to normal operation.
1.3.4 Unit 15 remains out of service.  Its four year overhaul started on 01 May.
1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore fish ladder.  The units are placed on standby, as necessary, during nighttime hours to float debris. 

1.5 Gatewell drawdowns are measured on Wednesdays; VBS drawdowns are measured daily on modified units.
2 ACTIVITIES

2.1 AWS Closures.  No AWS valves were closed.
2.2 STS/VBS Inspections.  STSs/VBSs inspections are scheduled for May.
Table 1.  STS and VBS inspections.

	Unit
	Date
	Run Hours
	Comments

	11-18
	NA
	NA
	NA


2.3 Dewatering and Fish Salvage.  No dewatering or salvage activities this week.
3 RESEARCH
3.1 Battelle kelt passage.  Monitoring with hydroacoustics near B2CC entrance to study kelt passage.  
3.2 Battelle spillway survival.  Installed equipment in the spillway for a spillway survival evaluation.
3.3 Fish Field Unit.  Sea lion observations continue.

3.4 NOAA Lamprey.  Construction of the lamprey ramp in the WA shore AWS began in mid January.
3.5 PSMFC Smolt Monitoring.  Five steelhead and 1 adult salmon crossed the separator bars.   2622 clipped yearling chinook were collected for NOAA Fisheries and Battelle. 
3.6 University of Idaho.  Spring chinook evaluations continue. 
3.7 USACE TDG.  New TDG monitors were installed in the spillway tailrace.

3.8 USDA Wildlife Services.  Hazing sea lions from the powerhouses continues. 

3.9 USGS.  Lamprey held in the FERL for tagging evaluation.
3.10 WDFW Fish Counting.  Visual counting continues Bradford Island and Washington Shore
3.11 WDFW/ODFW/CRITFC Sea lion hazing.  Hazing from boats continues.
4 FISHWAYS 

4.1 Project biologists inspected on 06 through 12 May.  
4.2  Fish Passage Plan Observations.
Table 2.  FPP Out of Criteria items.

	DATE
	LOCATion
	FPP itEMS
	CAUSE
	RESPONSE
	TR#

	BRADFORD ISLAND AND BRANCHES

	Once
	Calibration
	Check not done
	Short-handed
	New bio hired.  She starts on 14 May.
	

	CASCADES ISLAND

	All week
	FG6-10
	Closed
	Shaft and motor have parted ways.
	TR submitted.  See 4.2.6
	07-37350

	All week
	FG6-14
	½ way open
	Electricians need to do some rewiring.
	Electricians and riggers working out scheduling.
	

	WASHINGTON SHORE

	5/6-5/10
	Velocity meter
	Not reading at all.
	Improper wiring.
	Electrians repaired on 11 May.
	07-36771

	All week
	11A-S
	Closed
	Broken shaft
	Mech. working on it.
	07-40917

	5/6, 
	Weir 38
	Reading <1.0’
	Unknown
	Valve #2 adjusted.
	

	5/9
	Weir 38
	Reading >1.0’
	Early shad passage 
	No action taken.
	

	5/9
	Weir 37
	Reading >1.0’
	Early shad passage 
	No action taken.
	


4.2.1 PH1 electronic velocity meter is in an eddy.  Due to this placement, the velocity meter reads less than 1.5 fps.  Readings to meet FPP criteria are taken from the south end mechanical meter.

4.2.2 PH1 PCC diffusers FG2-22A and FG2-19 are closed or partially open.  Repairs scheduled for 07/08.  

4.2.3 FG3-3 stuck partially open.  Repairs are scheduled for winter 07/08.  TR#07-2660

4.2.4 FG3-5 has a damaged shaft and the valve won’t operate.  Repairs are scheduled for winter 07/08.  

4.2.5 FG3-6 stuck partially open.  Repairs are scheduled for winter 07/08.  

4.2.6 FG6-10 closed.  Only out of criteria when tailwater is above 19.0’ msl.  Mechanics will make repairs as soon as they finish with work on U2.

4.3 Fishway Temperatures.  Measurements are taken three weirs down from the count station at Bradford Island and at the junction of the UMT and the main ladder at Washington shore. Temperatures are in Fahrenheit.
Table 3.  Bonneville Fishway Temperatures.

	Date
	WS average
	BI average

	5/6
	54.0
	53.8

	5/7
	54.6
	54.6

	5/8
	55.5
	55.4

	5/9
	55.5
	55.3

	5/10
	55.3
	55.2

	5/11
	55.5
	55.4

	5/12
	55.7
	55.5


4.4 Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week.  

4.5 Adult Fishways. 
4.5.1 Calibration will be checked on weekends.
4.5.2 B-Branch and Cascades Island PLCs are in automatic.
4.5.3 FV3-7 is in manual until the A-Branch diffusers are repaired and operating correctly.
4.5.4 SLEDS installed at all main fish entrances. 

4.5.5 Stab plates remain installed in the FOGs due to the sea lion barriers not fitting properly.
4.6 Juvenile Fishways.
4.6.1 PH1 chaingates 1A and 1B remain closed at this time.
4.6.2 DSM1 continues to remain OOS.
4.6.3 DSM2 orifice lights continue to be lighted with the halogen bulbs.
4.7 Lamprey Fishways.  
4.7.1 The Bradford Island lamprey ramp remains in service as of 23 March.
5 CONSTRUCTION
5.1 B1 ITS Wall Removal.  Plans are to remove the wall, slope the floor, and install three automated gates.
5.2 PH1 STS/VBS removal and TIES removal.  Six TIES and 23PH1 STSs/VBSs will be scrapped.

5.3 PH2 VBS mods.  All units are complete.  Units 14 and 18 are scheduled for winter 07/08.
6 HAZMAT, SPILLS AND CLEANUP 
6.1 Gear oil from Gantry 7 got into gate slot 12C.  Orifices were closed from 1310 until 1510 to keep oil from entering the DSM2.  The amount in the gatewell was determined to be less than 1 quart and was cleaned up with absorbents.  A slight sheen was still present, so booms were left in place to absorb residual oil.
7 TELETYPES AND OTHER NOTEWORTHY ITEMS
7.1 A teletype sent on 22 December details tailwater operations for chum incubation.   Effective as soon as possible, a hard tailwater constraint to operate project tailwater no lower than 13.0 feet during all hours.  One pulse per day with a minimum duration of 20 minutes and minimum elevation of 13.5 feet. 
7.2 A teletype sent on 06 April requests juvenile spill start at 0001 on 10 April.  Bonneville will spill 100K 24 hours a day.  The spill table may be found in the 1007 FPP page Bon-51, Table Bon-15.  At no time will the project voluntarily spill above the 120% gas cap.
7.3 A teletype sent on 11 May requests the Project maintain a forebay soft constraint of 71.5’ to 72.5’ from 0900 until 1800 on 19-20 May.  This request is to allow a work party to stockpile debris on the kite board staging area, instead of leaving it strewn about the area.

JAMES R. MAHAR, P.E.

Operations Manager
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